Author's response to reviews

Title: Methylenetetrahydrofolate reductase gene variant (MTHFR C677T) and migraine: a case control study and meta-analysis

Authors:

Zainab Samaan (samaanz@mcmaster.ca)
Daria Gaysina (d.gaysina@nshd.mrc.ac.uk)
Sarah Cohen-Woods (sarah.cohen@iop.kcl.ac.uk)
Nick Craddock (craddockn@cardiff.ac.uk)
Lisa Jones (l.a.jones@bham.ac.uk)
Ania Korszun (a.korszun@qmul.ac.uk)
Mike Owen (OwenMJ@cf.ac.uk)
Andrew Mente (andrew.mente@phri.ca)
Peter McGuffin (peter.mcguiffin@kcl.ac.uk)
Anne Farmer (anne.farmer@kcl.ac.uk)

Version: 4 Date: 7 May 2011

Author's response to reviews:

Methylenetetrahydrofolate Reductase Gene Variant (MTHFR C677T) and Migraine: A Case Control Study and Meta-analysis
Samaan et al

Response to Reviewer’s comments
We have the pleasure in resubmitting our revised manuscript “Methylenetetrahydrofolate Reductase Gene Variant (MTHFR C677T) and Migraine: A Case Control Study and Meta-analysis” for further consideration for publication in BMC Neurology.

We would like to thank the reviewer again for revising the updated manuscript and the helpful comments. We have revised the manuscript incorporating recent comments.

More specifically we have added sentences to explain that the increased risk for MA in the presence of the TT genotype is limited to a small number of individuals who are carrying this variant. We also added that the findings of association between this gene variant and total migraine is most likely attributed to MA and not MO. All changes are highlighted in the text.

Thank you again for the helpful input.

Sincerely,
Dr. Zena Samaan, on behalf of co authors